Author's response to reviews

Title: Diagnosis of pericardial cysts using diffusion weighted MRI: a case series

Authors:

Asam Raja (araja@sbgh.mb.ca)
Jonathan Walker (jwalker@sbrc.ca)
Maneesh Sud (umsudm@cc.umanitoba.ca)
Joe Du (xjoedu@yahoo.ca)
Matthew Zeglinski (mzeglinski@sbrc.ca)
Andrew Czarnecki (andrewczarnecki@shaw.ca)
Negareh Mousavi (negarehm@yahoo.com)
Davinder S Jassal (djassal@sbgh.mb.ca)
Iain D Kirkpatrick (ikirkpatrick@sbgh.mb.ca)

Version: 5  Date: 12 August 2011

Author's response to reviews: see over
August 12, 2011

ATTENTION: PROFESSOR MICHAEL KIDD
EDITOR-IN-CHIEF
JOURNAL OF MEDICAL CASE REPORTS

RE: REVISED MS 1707198248525635: DIAGNOSIS OF PERICARDIAL CYSTS USING DIFFUSION WEIGHTED MRI: A CASE SERIES

Dear Dr. Kidd:

Thank you for your favorable review of our manuscript. Attached you will find a detailed response to the comments in addition to a revised manuscript with changes in bold. All authors have read and approved submission of the revised manuscript. We have no conflicts to declare. I will be the corresponding author.

Thank you for consideration of this revised manuscript.

Yours sincerely,

Davinder S. Jassal, MD, FACC, FRCPC
Associate Professor of Cardiology, Radiology and Physiology
Rm Y3010, Bergen Cardiac Care Centre
Section of Cardiology, Department of Internal Medicine
St. Boniface General Hospital
409 Tache Avenue, Winnipeg, Manitoba, Canada, R2H 2A6
204-237-2023
djassal@sbgh.mb.ca
Deputy Editor Comments:

1. “The editorial team would like to request the authors to amend the title. Please include the study design in your title.”

As suggested, the title of the manuscript has been changed to “Diagnosis of pericardial cysts using diffusion weighted MRI: a case series.”